Dear editor:
The authors summarized an interesting case with an unusual treatment method. I have some suggestions to improve the manuscript. The long versions of the abbreviations in the case report section must be given before the abbreviation use. The discussion part should be summarized. First, the HIFU and the general features of the mentioned tumor must be told. Then the unique aspects of the case should be compared with the literature. Other unnecessary details should be omitted. Figure legends are too sloppy. The radiologic explanations must be corrected. The subtitles of the figures (A, B, C etc) are not consistent with the explanations. The important features should be shown with markings, e.g. arrows, circles etc.
As requested, we revised the paper.

1. Abbreviations
2. Describe the general characteristics of HIFU and spindle cell tumor, and compare and summarize the uniqueness of this case with the literature.
3. Re-create pictures and picture descriptions.

Best regards,

All authors